Author's response to reviews

Title: Novel curcumin- and emodin-related compounds identified by in silico 2D/3D conformer screening induce apoptosis in tumor cells

Authors:

Melanie Fullbeck (melanie.fuellbeck@charite.de)
Xiaohua Huang (xiaohua.huang@charite.de)
Renate Dumdey (renate.dumdey@charite.de)
Cornelius Frommel (cornelius.froemmel@charite.de)
Wolfgang Dubiel (wolfgang.dubiel@charite.de)
Robert Preissner (robert.preissner@charite.de)

Version: 3 Date: 28 July 2005

Author's response to reviews: see over
To: The BioMed Central Editorial Team
editorial@biomedcentral.com

From: Melanie Füllbeck
Institute of Biochemistry
Charité-Universitätsmedizin Berlin
Monbijoustr. 2
10117 Berlin
Germany
Tel ++4930-450-528375
Fax ++4930-450-528942
E-mail: melanie.fuellbeck@charite.de

Berlin, 28th July 2005

Re: MS: 2137259566668425 - Novel curcumin- and emodin-related compounds identified by in silico 2D/3D conformer screening induce apoptosis in tumor cells

Dear Editor,

thank you for accepting my manuscript for publication in BMC Cancer.
Enclosed please find our final manuscript entitled “Novel curcumin- and emodin-related compounds identified by in silico 2D/3D conformer screening induce apoptosis in tumor cells”.

The required formatting changes regarding the references have been made. There were no further changes demanded from the reviewer.

Yours sincerely

Melanie Füllbeck